Author's response to reviews

Title: The association between foot-care self efficacy beliefs and actual foot-care behaviour in people with peripheral neuropathy: a cross-sectional study

Authors:

Byron M Perrin (b.perrin@latrobe.edu.au)
Hal Swerissen (h.swerissen@latrobe.edu.au)
Craig Payne (c.payne@latrobe.edu.au)

Version: 6 Date: 1 February 2009

Author's response to reviews:

We have made one, final change: Reference 14 contained 5 words in the author list that should not have been there.

The author list of reference 14 has now been fixed.

All other parts of the manuscript have been checked and there are no other issues.

Please note: At the end of the first sentence of the second paragraph page 8 I have re-typed the word "Additional". I did the same at the end of the fourth sentence of the third paragraph of page 8. This pertains to the figures but I am not sure if it needs to be a hyperlink or something like that rather than text.